

Guide to Best Practices in Faculty Mentoring

Office of the Provost



CONTENTS

Acknowledgments	3
Introduction	4
Letter from the Provost	5
Part I - Mentoring Overview	6
Introduction	6
Rationale and Definitions	6
General Principles	7
Mentor Roles	7
Mentoring Foci	8
Sponsorship vs. Mentorship	8
Formal vs. Informal Mentoring Relationships	10
Characteristics of Effective Mentoring	11
Mentoring Continuum	11
Qualities of Effective Mentors	11
Qualities of Responsible Mentees	12
Considerations for Effective Mentorship across Gender/Race/Ethnicity/Sexual Orientation/	
Culture/Religion	12
Part II - Organizing a Mentoring Program: Best Practices	14
Establishing School/Departmental Mentoring Program	14
Mentor/Mentee Pairing	15
Responsibilities	15
Frequency of Mentoring Meetings	15
Faculty Group Mentoring	15
Interdisciplinary/Team Mentoring	15
Establishing Oversight, Evaluation, and Sustainability	16
Part III - Tools for Mentors	17
Part IV - Strategies for Mentees	18
Mentoring (Developmental) Networks	19
Appendices	20
Appendix A	20
Checklist: Best Practices for Schools/Departments, Mentors, and Mentees	20
For Schools/Departments	20
For Mentors	21
For Mentees	21
Appendix B	22
Logic Model for Schools/Academic Units Implementing Mentoring Programs	22
References	23

ACKNOWLEDGMENTS

The fifteen members of the Provost's Advisory Council for the Enhancement of Faculty Diversity provided leadership and counsel in creating this guide. Individually, each possesses a deep commitment to advancing faculty diversity at Columbia; collectively, they represent the enormous breadth of Columbia's academic community. We particularly note the invaluable contributions of our Climate Change Working Group:

Anne L. Taylor, Chair, Working Group; Office of the Executive Vice President and Dean of the Faculties of Health Sciences and Medicine;

Office of the Chief Executive, Columbia University Medical Center

Katherine W. Phillips

Office of the Dean, Columbia Business School

Ruben L. Gonzalez Jr.

Department of Chemistry, Faculty of Arts and Sciences

Dennis A. Mitchell

Office of the Provost;
Office of the Dean and Division of
Community Health, College of Dental
Medicine

Samuel K. Roberts

Department of History,
Faculty of Arts and Sciences;
Department of Sociomedical Sciences,
Mailman School of Public Health

We also want to give special thanks for the leadership and support of **Lee Goldman**, Executive Vice President and Dean of the Faculties of Health Sciences and Medicine and Chief Executive of Columbia University Medical Center, as well as the staff members of the Office of the Senior Vice President for Faculty Affairs and Career Development at Columbia University Medical Center and the Office of the Provost, who played a critical role in researching and assembling resources.

Clara Lapiner

Director of Faculty Professional Development, Columbia University Medical Center

Shana L. Lassiter

Assistant Provost for Faculty Diversity and Inclusion, Columbia University

Adina Berrios Brooks

Director for Faculty Diversity and Inclusion, Columbia University

This guide is a result of a review of relevant literature and discussions with the Climate Change Working Group as well as with deans and senior and junior faculty throughout the University. While these discussions were focused on the mentorship needs of diverse faculty, it was recognized that attention to the mentoring needs of all faculty would substantially benefit the university. Thus, this guide provides information on the principles of mentoring, with attention to the nuances of mentoring important to diverse faculty.

First printing, August 2016.

INTRODUCTION

The Offices of the Senior Vice President for Faculty Affairs and Career Development at CUMC and the Vice Provost for Faculty Diversity and Inclusion developed this guide in collaboration with the Provost's Advisory Council for the Enhancement of Faculty Diversity. This guide was developed to help academic leaders and faculty members who wish to use mentoring as a strategy to facilitate faculty success. Specifically, this guide was written to inform three key constituencies:

- Schools/departments wishing to implement formalized mentoring programs for faculty
- Faculty who wish to act as mentors to junior faculty, postdoctoral fellows, and graduate students
- Faculty who need or wish career guidance in addition to, or in the absence of, formal school/departmental mentoring programs

These basic principles of effective mentoring may be utilized by schools, departments, and individual faculty to provide guidance to junior faculty, in formats most suited to the needs and culture of the academic unit as well as to the individual. This guide pays particular attention to the nuances of mentoring of particular importance to women and diverse faculty. Finally, the guide offers guidance to faculty who may wish to broaden and increase their mentoring opportunities and networks.

A note on organization:

- Part I provides an overview of principles of mentoring derived from a literature review
- Part II provides strategies and best practices for academic units wishing to organize different types of mentoring programs for faculty
- Part III provides guidance for those committed to being mentors
- Part IV provides strategies and best practices for mentees who are committed to identifying and optimizing their mentoring relationships

This guide is written as a living document; in that spirit, we invite feedback on what works, what doesn't, and what should be added. Please address all comments and suggestions to Vice Provost for Faculty Diversity and Inclusion Dennis A. Mitchell at dmitchell@columbia.edu.

LETTER FROM THE PROVOST



Dear Colleague:

Columbia University aspires to be the go-to institution for the world's greatest scholars. We cannot achieve this without realizing our core values of both inclusion and excellence. This requires sustained focus on equity in all of our efforts to recruit, hire, promote, and retain an exceptionally well-qualified faculty. The University's *Guide to Best Practices in Faculty Search and Hiring*, first released electronically in 2014, provides guidance for our recruitment and hiring efforts. The current *Guide to Best Practices in Faculty Mentoring* serves as a companion resource, providing direction for promotion and retention efforts within each department or school.

This guide is intended to assist schools, departments, and faculty in implementing mentoring programs tailored to their needs. It does not replace existing University, school, or department procedures, but rather serves as a framework and supplemental resource. It provides an overview and guidance for schools and departments seeking to develop new mentoring programs, and serves as an asset for those schools and departments with robust mentoring programs already in place. The Guide also provides targeted resources for mentors and mentees, with a particular focus on empowering junior faculty members to seek and shape important mentoring relationships.

We developed this manual because we believe that adherence to its guidelines will have a positive impact on faculty success and will enhance the climate of inclusiveness University-wide, thus building a stronger University community. A thriving, diverse faculty is essential to creating a dynamic learning and working environment that will prepare all of our students to lead in our global society.

We hope you will find this resource valuable in your role as mentor, mentee, and/or champion of mentoring initiatives within your unit. Thank you for all that you do to strengthen our community and ensure the future excellence of Columbia University.

Sincerely,

John H. Coatsworth Provost

INTRODUCTION

Vital faculty are essential to Columbia University's mission to be one of the world's preeminent centers of research as well as a distinctive and distinguished learning environment. Mentoring is a highly successful, adaptable, and practical strategy for supporting faculty members' success and satisfaction across their career. Mentoring has been shown to enhance research productivity (Bland & Schmitz, 1986; Bland et al., 2002; Byrne & Keefe, 2002), to enhance teaching effectiveness (Williams, 1991), and to increase faculty retention, recruitment, productivity and satisfaction, as well as to decrease faculty attrition. In addition, mentoring may promote a more positive organizational climate (Corcoran & Clark, 1984; Melicher, 2000).

RATIONALE AND DEFINITIONS

Definitions, concepts, and practices of mentorship for university faculty vary considerably, but all place value on a relational structure that provides specific knowledge required for faculty to develop successful independent scholarship, as well as essential professional relationships that place faculty within the network of scholars in their discipline. These are often framed as career advancing or "instrumental" functions or as psychosocial or "expressive functions," but they are closely related and overlapping (Zellers, Howard, & Barcic, 2008; Chao, Walz, & Gardner, 1992; Mott, 2002). Though most frequently conceptualized as a single, formalized, dyadic, hierarchical relationship between a senior and junior faculty member (Zellers, Howard, & Barcic, 2008), more recent concepts of mentoring would suggest that effective faculty mentoring should involve both formal and informal relationships with a broad array of professional colleagues (mentoring networks, constellations, distributed mentoring, partnerships) (Zellers et al., 2008; Pololi & Knight, 2005; De Janasz & Sullivan, 2004; Kram & Isabella, 1985; McClurken, 2009; Van Emmerik, 2004; Ibarra, 1993). Mentoring networks can provide guidance and multidimensional perspectives on developing contemporary academic careers, may reduce the power dynamics often present within the junior-senior pairing, and can include individuals within and outside of academic environments (Pololi & Knight, 2005; McClurken, 2009). It has been suggested that mentoring networks are of particular

importance to women and diverse faculty, who are less likely to find spontaneous dyadic mentoring relationships that address the full range of their career concerns (Mott, 2002; Zellers et al., 2008; Pololi & Knight, 2005; Van Emmerik, 2004; Chesler, Single, & Mikic, 2003; Chesler & Chesler, 2002; Bickel, 2014). Despite this, the development of effective mentoring networks may still be more difficult for diverse faculty (Ibarra, 1993). Further, the "mentoring network" model empowers junior faculty to actively identify and set in motion essential mentoring relationships suited to their unique needs, as opposed to more passive dependency on department or senior faculty to initiate mentorship. Because successful mentoring relationships require active and committed engagement on the part of both mentor and mentee, an essential concept for junior faculty to fully embrace is that they (mentees) are empowered to seek and shape important mentoring relationships (McClurken, 2009; Pololi & Knight, 2005; Fleming et al., 2015; Chesler & Chesler, 2002; Chesler et al., 2003; Bickel, 2014).

Specific points of general knowledge of importance (i.e., requirements for academic promotions, specific skills) may be efficiently provided to groups of faculty via seminar or workshop format. Often categorized as "group mentoring," it is an efficient way to be sure that a common fund of important career development information is delivered to all junior faculty, particularly closing the "information gap" experienced by women and underrepresented minority faculty. Finally, the focus of faculty mentoring has broadened to include faculty beyond early-career stages, including midlevel and senior faculty who face career transitions; these faculty may utilize self-initiated "peer or near peer" mentoring relationships to a much greater extent than junior faculty, but may benefit from specific school/departmental facilitated group or peer mentoring programs.

At the institutional or organizational level, mentoring has many benefits that increase organizational strength and productivity and establish an inclusive and supportive scholarly community (Zellers et al., 2008; Boyle & Boice, 1998). Mentoring has been shown to enhance research productivity (Bland & Schmitz, 1986, Bland et al., 2002; Byrne & Keefe, 2002) and teaching effectiveness (Williams, 1991; McMurtrie, 2014; Zellers et al., 2008); and has

been shown to increase faculty retention, recruitment, productivity, and satisfaction, as well as to decrease faculty attrition, and to promote a more positive organizational climate (Corcoran & Clark, 1984; Melicher, 2000; Zellers et al., 2008). Thus, fostering a culture to support the success of its faculty is in the best interests of an institution.

GENERAL PRINCIPLES

Mentoring is commonly understood to imply a supportive relationship designed to guide the successful integration of new members into an organization and to enhance the subsequent ability of those members to add value to the organization (Bland et al., 2009; Mott, 2002; Zellers et al., 2008). In this context, we wish to precisely define the important principles of these types of relationships as they relate to university faculty:

- 1. Mentorship is a collaborative learning process that draws upon the knowledge of a variety of faculty who can provide guidance (senior faculty, near peers, and peers may all function as mentors) (Kram & Isabella, 1985) to new faculty entering the professoriate or to more senior faculty transitioning to new roles. The relationship has been described as a "reciprocal, supportive, and creative partnership of equals" (Mott, 2002; Kram & Isabella, 1985). Mentoring requires active committed engagement on the part of both mentor and mentee. It is dependent upon the willingness of those acting as mentors to invest time for guidance on an ongoing basis. It also requires the commitment of those needing guidance to actively identify specific developmental goals and to invest time and energy to achieve them.
- 2. Mentoring should help junior faculty successfully acquire the key competencies (scholarly independence, educational skills, and preparation for academic advancement), as well as the constructive professional relationships (professional networks) within the institution and beyond needed to develop a productive career (Zellers et al., 2008; Bhagia & Joyce, 2000; Chao et al., 1992).
- 3. The traditional, hierarchical, dyadic mentoring relationships may be enriched by an additional network of individuals providing very specific guidance in areas of professional development that may not be addressed within a single dyadic relationship (Zellers et al., 2008;

- Kram & Isabella, 1985). Formal, departmentally assigned mentoring relationonships and informal, mentee-initiated relationships may be complementary and support different aspects of career guidance. Mentoring networks are of particular relevance to faculty groups traditionally underrepresented in the professoriate (Pololi & Knight, 2005; Pololi, 2013; Bickel, 2014; Chesler & Chesler, 2002; De Janasz et al., 2003; Van Emmerik, 2004).
- 4. Mentoring need not be limited to junior faculty, as midlevel and senior faculty may wish to focus on career transitions and new directions. The structure of these relationships may be less formalized and more dependent on peer or near peer relationships; professional "coaching" and grouptaught skills acquisition, through facilitation by school/ department, may be useful.
- 5. Mentoring relationships should evolve over time and may focus on one or several elements required for career success depending upon the career stage of the mentee, the career goals to be met, the level of guidance required, and the nature of the input from mentors. Thus, they may be enduring, long-term relationships that evolve over time into collegial rather than mentoring relationships, or they may be more transient relationships focused on specific areas of guidance at key career points.

MENTOR ROLES

Part of the variability in the definition of mentoring derives from the fact that mentors are perceived as having many roles that may be grouped into two broad categories: "technical or instrumental career functions" and "psychosocial or expressive career functions" (Zellers et al., 2008; Chao, Waltz, & Gardner, 1992; lbarra, 1993; Mott, 2002).

Technical or instrumental career functions

- Advisor for overall professional goals and career choices
- Advisor for development of academic scholarship
- Facilitator of professional networking within and outside of the institution
- Sponsor who provides specific strategic opportunities with career relevance
- Advisor for the development of teaching skills

Psychosocial or expressive career functions

- Promoter of scholarly values and professional integrity
- Advocate
- Role model
- Coach
- Intellectual challenger
- Colleague
- Supporter

Given the many roles that mentors may play and the greater diversity of the professoriate, it may be unrealistic to expect a single individual to perform all roles equally well. Thus, the value of the concept of multiple mentors and mentoring networks has become considerably more important in recent years (Mott, 2002; De Janasz & Sullivan, 2004). Additionally, because traditional mentoring relationships have often been the result of senior faculty gravitating to mentorship of junior faculty "like themselves," the technical and psychosocial functions may be intermixed. Identification and development of high quality mentoring relationships providing both instrumental and expressive functions have been much more challenging for women and members of underrepresented demographic groups (Mott, 2002; Pololi & Knight, 2005; Ibarra, 1993). Data strongly suggest that because these demographic groups are less likely to find traditional, dyadic mentoring relationships of high quality (Mott, 2002; Ibarra, 1993; Ibarra et al., 2010; Bickel, 2014; Daley, Wingard, & Reznik, 2006; Girves, Zepeda, & Gwathmey, 2005), they are less likely to progress in academia (Fang et al., 2000; Daley et al., 2006; Mark, 2001; Bickel, 2014). As the professoriate has grown more diverse, mentorship requires consideration of the social dimensions of career development particularly experienced by women and underrepresented minority faculty (Daley et al., 2006; Bland et al., 2009; Girves, Zepeda, & Gwathmey, 2005; Ibarra et al., 2010). These include cultural taxation, isolation, and exclusion from informal collegial networks, unintended bias, and devaluation of scholarship focused on minorities or women, and biculturalism (Turner, Myers, & Creswell, 1999). Thus, mentoring networks and multiple mentors specifically identified and sought by historically underrepresented groups may be of special value to guide management of the unique challenges experienced by these faculty groups.

MENTORING FOCI

The foci of faculty mentoring should include guidance in multiple domains of career development. These include, but may not be limited to the following:

- 1. Development of independent scholarship/research
- 2. Development of internal and external professional networks crucial to recognition as an independent scholar
- 3. Teaching skills
- 4. Development as a clinician or applied practitioner of a discipline
- 5. Strategies for success and advancement within the institution, school, or department, with attention paid to formal as well as informal measures of success
- 6. Requirements for academic advancement
- 7. Overall career planning, including short-, mid-, and longterm goals
- 8. Management of career challenges of particular relevance to women and underrepresented minority faculty
- 9. Sponsorship

SPONSORSHIP VS. MENTORSHIP

Mentorship and sponsorship both enhance career advancement, and mentors may also be sponsors; however, these two advisory functions are quite different. While mentors provide guidance, feedback, and psychosocial support on an ongoing basis, sponsors provide specific strategic opportunities to an individual at a particular time (Ibarra et al., 2010). Mentors can be peers, near peers, or more senior and provide ongoing guidance on technical as well as psychosocial functions. By contrast, sponsors are generally more senior in rank with sufficient influence in their field to provide key opportunities, but may have no other supportive function (Ibarra et al., 2010). Actions such as advocating at a critical time on behalf of the faculty, nominating faculty for selected awards or important organizational memberships, or making strategic introductions to key people within the discipline of the scholar can be single or episodic sponsoring actions, separate from mentoring, that nonetheless may have very significant career impact.

Though the roles of a sponsor and a mentor are different in function, a mentor can often be a sponsor if the mentor in question has the necessary authority, connections, and network to provide key opportunities (lbarra et al., 2010; Zellers et al., 2008). However, a mentor and sponsor may not always be found in the same person. Both mentorship and sponsorship are influential to career success, and have been shown to be particularly important for developing the "social capital" of women and minority faculty in academia (lbarra et al., 2010; Bickel, 2014). A 2008 survey of men and women in business companies found that both men and women had mentors, but the quality of mentorship greatly varied between men and women (lbarra et al., 2010). As Hermnia lbarra, Nancy Carter, and Christine Silva wrote:

"All mentoring is not created equal, we discovered. There is a special kind of relationship—called sponsorship—in which the mentor goes beyond giving feedback and advice and uses his or her influence with senior executives to advocate.

for the mentee . . . Furthermore, without sponsorship, women not only are less likely than men to be appointed to top roles but may also be more reluctant to go for them (2010, p. 82)."

Because sponsorship may occur outside of traditional assigned mentoring dyads, and it is often initiated by a senior faculty member who identifies in some way with a particular junior faculty member, this quality of "sponsoring" may present a major obstacle for women and minority faculty (Bickel, 2014), who differ in many ways from established senior faculty (Chesler & Chesler, 2002; Chesler et al., 2003). This was famously illustrated by Sheryl Sandberg, author of *Lean In*, who stated that "It's wonderful when senior men mentor women. It's even better when they champion and sponsor them. Any male leader who is serious about moving toward a more equal world can make this a priority and be part of the solution. (2013, p. 71)."

MENTORING MODELS						
	TRADITIONAL	PEER AND NEAR PEER	GROUP	MENTORING NETWORKS	SPONSORSHIP	
STRUCTURE	Hierarchical Often assigned, highly structured	Peer, near peer Often mentee initiated, may be facilitated by school/department	Hierarchical Often time limited	May include any level of faculty. May be mentee or mentor initiated or facilitated by school/department	Hierarchical- highly influential senior faculty who provide key opportunities	
FORMAT	One mentee with one mentor or team of mentors	One-to-one or small collaborative group Nonhierarchical power dynamic Reciprocal information sharing and psychosocial support	One or small number of mentors for specific skills, training, or information distribution	Multiple relationships with a variety of advisors. Variable duration, scope, and nature of support	Can be single or intermittent actions not associated with ongoing mentorship	
TYPICAL CAREER STAGE OF MENTEE	Early	All career stages	All career stages	All career stages	Junior	
TYPICAL CAREER STAGE OF MENTOR	Mid or senior	All career stages	All career stages	All career stages	Senior	

FORMAL VS. INFORMAL MENTORING RELATIONSHIPS

Formal mentorship is defined as a structured and assigned pairing of mentor and mentee with clear goals, meeting structure, duration, and definition of target outcomes. A meta-analysis of studies on the career benefits associated with mentoring and sponsorship found that, overall, individuals who were mentored and sponsored reported better objective career outcomes (compensation and number of promotions) as well as better subjective outcomes (career satisfaction, career commitment, and job satisfaction) (Allen et al., 2004).

In a study of two mentoring programs for junior faculty, formal mentoring was deemed to be, overall, more effective than informal or "natural" mentoring (Boyle & Boice, 1998). The study showed that mentoring relationships continued longer and were better sustained in the formal program. Additionally, the sustainability of the mentoring relationship was less dependent on personality, but instead more on specific tasks to which the mentoring pair attended (Boyle & Boice, 1998). Research productivity has been found to be increased by formal mentorship. A study of medical school faculty found that having a formally designated mentor was the second-best predictor of higher research productivity (Bland, Seaquist, Pacala, Center, & Finstad, 2002).

Other literature directly comparing formal structured mentorship to informal mentorship surprisingly found informal mentorship to be more positively correlated with career-related support, organizational socialization, and satisfaction than formal mentorship (Chao et al., 1992). However, this study measured overall career satisfaction of alumni from a single university and thus did not measure the specific outcomes required of university faculty.

A significant risk in traditional dyadic pairings is the inherent power differential between senior and junior faculty and the pressure to replicate the senior generation rather than promote the emergence of newer scholarly approaches, fields, or concepts of professorial roles. This has been described as "academic cloning" (Zellers et al., 2008; Tenner, 2004).

Mismatches, either personal or professional, between assigned mentor-mentee pairs can occur with negative and potentially disruptive consequences to career progression (Mott, 2002). There are, however, data demonstrating that formal mentorship is likely to be more enduring and successful when the mentee and mentor(s) have some choice in the assignment (Zellers et al., 2008). However, even when the mentor and mentee have a choice in pairing, such pairings may not be successful, and thus some oversight with the authority to dissolve and reassign is required by the school/department sponsoring the formal assignment.

Informal mentorship, by contrast, occurs as a result of spontaneous initiation by either the mentee or mentor based on perceived commonality in some aspects of career goals (Zellers et al., 2008). These mentoring relationships can be highly effective as they evolve based on mutual affinity, though they may lack the regularity of communication and the comprehensive approach to multiple dimensions of career development that occurs in structured formal programs. However, informal mentoring relationships may be of great value in providing additional perspectives on scholarship, as well as psychosocial support. Other aspects of career development centered around gender, race, or ethnicity may not be addressed within a formally assigned mentoring relationship. Such discussions may be more easily addressed in informal, self-initiated relationships. In addition, the "self-assignment" in informal relationships lessens the likelihood of unsuccessful or negative mentormentee matches. In particular, peer and near peer mentoring relationships remove the inherent senior-junior power dynamic defined by roles and rank, and may permit freer reciprocal exploration of career issues (Mott, 2002). Such relationships may be particularly important to women and underrepresented minority faculty. The literature comparing formal and informal mentoring points to advantages and disadvantages of both. Thus, junior faculty are well advised to not limit mentorship relationships to either format, but to use both formats to their advantage (Chao et al., 1992).

MENTORING CONTINUUM

Supervision

Informal participantinitiated Mentoring Enhanced
Informal
Mentoring
(facilitated but
participant
driven)

Formal Mentoring (department or school run)

Sponsorship

Professional Coaching

Adopted from Harold Pincus, MD, Department of Psychiatry, Columbia University College of Physicians and Surgeons

CHARACTERISTICS OF EFFECTIVE MENTORING

In formal mentoring relationships, the roles and responsibilities of mentors and mentees should be defined and made explicit from the outset. Both the mentor and mentee have responsibilities for maintaining the productivity of the relationship.

The literature distills characteristics of effective mentoring to include the following:

- 1. Development of clear, agreed-upon career goals and plans to achieve them:
 - a. Objectives
 - b. Strategies for achieving them
 - c. Specific deliverables
 - d. Timelines
- Agreeing upon roles for each mentor when there is more than one mentor (particularly important for interdisciplinary scholars) as well as whether technical or psychosocial functions are the focus
- 3. Setting clear rules regarding meetings (frequency, agenda, and deliverables)
- 4. Establishing accountability and oversight for mentors and mentees when formal assignment is made
- Maintaining confidentiality when requested and appropriate
- 6. Agreeing on communication mechanisms, frequency, and style
- 7. Measuring progress with timelines
- 8. Providing constructive feedback

- 9. Creating an environment of trust and open communication
- Gradually reducing level of guidance, with encouragement toward academic independence

MENTORING CONTINUUM

As mentoring relationships evolve, so do the roles and functions of the members in the relationships. The figure above demonstrates the ways in which roles can vary over time and in different forms of mentoring. Mentoring can vary in the proportion of directive versus nondirective advising on the part of the mentors, depending on the stage of development of the mentee. While close supervision may be appropriate for student mentees, faculty mentor/mentee relationships are between individuals who also are professional colleagues and thus should be collegial and collaborative rather than highly directive.

Qualities of Effective Mentors

- 1. Commitment to the time required for meeting and advising
- 2. Commitment to reviewing the mentee's work
- Guidance on setting realistic career vision/goals/ objectives
- 4. Collaborative development of appropriate goals and strategies for achieving goals
- 5. Guidance and resources for developing academic competencies, in scholarly research, teaching, scholarly presentations, and overall career management
- 6. Facilitation in the development of academic networks
- Careful and active listening, ability to work across boundaries of gender/race/ethnicity/sexual orientation/ culture/religion

Status of Women and Minorities in U.S. Degree-Granting Institutions, 2011

	Women	Men	Minorities*	
Full-Time Faculty	44%	56%	19%	
Assistant Professor	49%	51%	23%	
Associate Professor	42%	58%	20%	
Full Professor	29%	71%	15%	

From U.S. Department of Education, National Center for Education Statistics, 2011 national study of full-time instructional faculty in degree-granting postsecondary institutions

- 8. Provision of constructive feedback
- 9. Attention to measuring progress
- 10. Consideration of sponsorship when appropriate and possible
- 11. Maintenance of confidentiality when appropriate and creation of an environment of trust
- 12. Respect for personal boundaries

Qualities of Responsible Mentees

- 1. Active engagement in the identification of specific developmental goals for which guidance is sought
- 2. Commitment to setting meeting times and agendas to ensure a productive discussion
- 3. Initiative and follow through in both setting and completing agreed-upon goals
- 4. Commitment to hard work and integrity
- Preparation of agreed-upon deliverables in adequate time for review by mentors, with respect for the time constraints of mentors
- 6. Development of listening skills and willingness to work outside of "comfort zones," across boundaries of gender/race/ethnicity/sexual orientation/culture/religion
- 7. Ability to solicit and consider thoughtful feedback, both positive and negative
- 8. Active commitment to developing scholarly independence
- 9. Respect for personal boundaries

CONSIDERATIONS FOR EFFECTIVE MENTORSHIP ACROSS GENDER/RACE/ ETHNICITY/SEXUAL ORIENTATION/ CULTURE/RELIGION

Though women start academic careers in almost equal numbers as men, the numerical difference between the groups widens with advancing rank, so that men lead with slightly greater than 2:1 ratio at the full professor rank (Snyder, 2015). Snyder (2015) cites that in 2013, the National Center for Education Statistics, as part of a U.S. Department of Education report, published data from 2011 showing that women and men entered higher education faculty positions in similar numbers at the assistant professor level, but that as they move up the ranks, the quantitative difference between men and women changes dramatically, with significantly more men than women at the full professor rank. Minorities, constituting 19 percent of all full-time faculty, followed a similar pattern as women (see table above) (Snyder, 2015).

Data show that women and minorities are less likely to be promoted and are more likely to stay at the same rank longer (Fang et al., 2000; Alexander & Lang, 2008; Liu & Alexander, 2010; Palepu et al., 1998), which can lead to higher attrition among these groups (Cropsey et al., 2008; Mahoney et al., 2008; Nunez-Smith et al., 2007).

Other studies have echoed these findings and have attributed this "leaking pipeline" for women and minority faculty to subtle cumulative barriers (lbarra et al., 2013; lbarra et al., 2010; Fried et al., 1996; Smith et al., 2015)

^{*} Includes Black, Hispanic, Asian, Pacific Islander, American Indian/Alaska Native, and of two or more races

including unconscious bias, lack of institutional support, feelings of isolation, poorer-quality mentoring relationships, absence of sponsorship, work/life imbalance, and cultural, societal, and institutional norms. For women in particular, barriers such as inflexible and unsupportive work/life policies (i.e., tenure-clock systems that are not flexible around issues of child or elder care, or being required to work outside of traditional hours), lack of access to sponsorship or peer networks, and poorer-quality mentoring relationships have been identified as contributing to a cumulative disadvantage and lower perceived capabilities of women to achieve their goals (Ibarra, 1993; Fried et al., 1996; Smith et al., 2015; Pololi et al., 2013). While overt discriminatory practices have diminished over the years, women and minorities continue to face less overt but nonetheless substantive cumulative challenges (Ibarra et al., 2010; Ibarra et al., 2013; Girves et al., 2005; Daley et al., 2006). Mentoring relationships often form more spontaneously between individuals who easily identify important commonalities between themselves. This aspect of mentoring relationships may be particularly challenging for women and minorities whose mentors are likely different in a multitude of ways and who have faced different challenges (Bland et al., 2009; Zellers et al., 2008; Ibarra, 1993).

Women and underrepresented minority faculty experience unique social dimensions to academic career development that may very significantly impact professional advancement. In 2012, the provost of Columbia University commissioned a study to assess the needs of diverse junior faculty. Focus groups of twenty-two junior women and minority faculty were interviewed regarding their view of unique potential barriers to career success. Four themes were identified by the focus groups as having a negative impact on the success of women and minority faculty:

- disproportionate service burden
- feelings of isolation
- lack of mentorship
- lack of clarity about promotion and tenure process

These factors were quite congruent with those reported in the literature as listed below (Turner et al., 1999; Bland et al., 2009):

- Professional isolation, the experience of being the "lonely only" one of a particular demographic group within one's unit, with interactive challenges not experienced by senior or majority peer faculty.
- Exclusion from spontaneous or informal collegial networks, whether intentional or not.
- Unintended bias on the part of colleagues, as well as students and trainees, who may place lower value on scholarly contributions of women and underrepresented minorities.
- Societal norms about acceptable behaviors (i.e., expectations regarding gendered behavior) that may be at variance with professional behaviors required for successful advancement.
- Cultural taxation—the overburdening with service of value to the institution, but not of value for the career advancement of the faculty member.
- Biculturalism—the strain of being required to straddle or be fluent in more than one culture. This becomes particularly relevant when behaviors that are normative in one culture (i.e., modesty and humility) are at variance with behaviors expected for career success within academic disciplines (i.e., self-promotion and outspokenness).
- Devaluation of scholarship focused on issues highly relevant to women or underrepresented minorities.

Moreover, as pointed out in an excellent study and review of the literature (Williams et al., 2014), when there is the intersection of both race/ethnicity and gender, women of color report patterns of bias that incorporate both race/ethnicity and gender-based bias ("double jeopardy"). Of importance, such bias may focus more or less on gender compared to race/ethnicity such that the experience of women of color is not uniform and may significantly differ from that of white women.

Thus it is essential for mentors and mentees to be self-reflective and mindful regarding potential subtle barriers for women and underrepresented minority faculty and to maintain a focus on achievement of the specific tasks and goals that contribute to career success.

It is evident from the literature review that mentoring can have a positive impact on faculty career development as well as on departments and institutions as a whole. However, programs may vary in their effectiveness in meeting the needs of faculty and departments. Careful consideration of the structure, model, culture of the academic unit, and resources available to the unit—and devoted to the programs by the academic unit implementing a mentoring program—will ensure that faculty receive appropriate support to meet their goals and the goals of the schools/departments.

ESTABLISHING SCHOOL/ DEPARTMENTAL MENTORING PROGRAM

School/departmental programs should aim, at minimum, to assist all junior faculty to:

- 1. develop independent scholarship/research productivity
- 2. develop the necessary professional networks
- 3. identify determinants essential for academic advancement in that unit
- identify whether additional desired support should include proactive recognition and mitigation of factors that disproportionately negatively impact women and underrepresented minority faculty

Schools/departments should identify which mentoring model(s) will most effectively align with the needs of their faculty, with faculty demographics, and with the culture of the school/department considered. It would be wise for the department to clearly establish expectations regarding mentoring activities, as well as training for mentors. For example, a unit may choose to assign a principal mentor, but may also encourage or facilitate mentoring networks to bring together faculty of similar rank, interest, or demographics, or to provide group sessions covering information on advancement or particular skills.

Models

A combination of modalities may decrease the mentoring burden on individual senior faculty, as well as decrease the reliance of junior faculty on a single mentoring relationship:

- traditional hierarchical senior-junior faculty relationship (may be either a senior team or a senior individual)
- facilitation of groups of mentees matched to one or to a

- small group of senior mentors that are organized around the acquisition of specific skills or problem solving
- peer and near peer mentorship groups where faculty of a similar or near rank, or of similar demographics, meet individually or as a small collaborative group
- active consideration of appropriate sponsorship opportunities

Key Steps

In the process of identifying and creating a mentorship program model, the key steps outlined below may be useful:

- Consultation with junior and senior faculty: In order to build programs that are aligned with the needs of both junior and senior faculty and thus likely to be sustained, faculty should be involved in the process of building a program consistent with the needs and culture of the department. The scope of mentoring programs may be broad and include both career advancement and psychosocial support, or the academic unit may choose more narrowly focused mentoring. Clarity regarding expectations of the time commitment and efforts on the part of the mentors and mentees is essential. In addition, making sure that mentoring work is fairly distributed among senior faculty may require some training in mentoring skills (Pfund et al., 2006). It is also extremely important that all demographic groups of junior faculty (i.e., women, underrepresented minorities) have equivalent support and access to information supportive of career
- Identify the structure: Departments and schools should identify which models might be best suited for their faculty and for the needs of the department or school overall. For example, skills acquisition—such as principles of grant writing, teaching skills, research team management, and understanding the promotion process—lends itself to seminar or group formats, thereby allowing units to share resources to disseminate this information. By contrast, individual scholarly or research advising should be department specific since scholarship content and network development will be discipline specific.
- Resources: Departments and schools should identify what kind of resources or tools will be available to provide faculty mentors with needed skills and incentives. Attention should be paid to equitable distribution of the work of mentorship among senior faculty. Incentivizing faculty mentors in ways

such as making mentorship an integral part of the annual evaluation and promotion process—and finally, recognition and celebration of the importance, value, and the successes of excellent mentors through awards—helps to build a culture of mentorship. Resources for both mentors and mentees should also include tools for communications and/or programs that aim to enhance mentoring relationships.

MENTOR/MENTEE PAIRING

If units decide to assign mentor/mentee pairs for ongoing advising, pairs can be constituted in various ways. The options include assignment (either a mentoring committee or individual mentor) by the department, allowing mentors and mentees to fully self-select, or some combination of these two options. Data suggest that pairings may be more sustained and successful when both the mentor and mentee have some choice, but departments may also consider the scholarly basis for assignments. Departments should also be mindful of the data showing that women and underrepresented groups are less likely to develop high-quality mentoring relationships when self-selection is the sole method of assignment, and so should ensure that these faculty groups have equal access to excellent mentors.

RESPONSIBILITIES

Creation of specific annual goals, meeting schedules, deliverables, and measures of progress must be the joint work of mentors and mentees.

The mentee/junior faculty should be responsible for creating meeting agendas to include an update on interim activities related to their academic performance and career development. There should be discussion of progress in each career development domain and agreement concerning goals/strategies/deliverables for the next period. Additionally, the mentor should be sensitive and receptive to issues relating to sense of isolation, exclusion from school/ department, or disproportionate service burden that can be particularly experienced by underrepresented minority and women faculty. While it is the responsibility of the mentor to advocate for, advise, coach, and support the mentee, mentees are responsible for collaboratively developing appropriate goals and meeting expectations, and for the hard work and integrity necessary to develop academic independence.

FREQUENCY OF MENTORING MEETINGS

The frequency and level of detail of mentoring discussions may vary depending on the level of advancement of the junior faculty (i.e., it may be determined that more frequent meetings would be necessary for faculty in early years, with tapering of frequency or change in focus as faculty advance), as well as the level of independence of the junior faculty. An annual review of academic goals to assess overall progress is also advisable. The annual review may be clearly distinguished from mentoring meetings by its format and comprehensive review of all aspects of career development. Both the mentor and mentee should keep a record of the meetings as a measure of progress and for reference. The academic unit may decide whether or not it wishes to formalize review of these records to ensure that assigned mentoring meets the standard set by the school/department.

FACULTY GROUP MENTORING

In addition to school/departmental formal individual mentoring, academic units may wish to facilitate peer or near peer meetings in which groups of faculty may share their experiences, concerns, and problem-solving approaches. They may also provide opportunities for junior faculty networking with external scholars.

Other group mentoring meetings may enhance the knowledge and transparency of promotion policies and processes, departmental or school-specific academic expectations, and information on resources and opportunities available to faculty. Group meetings focusing on information critical to advancement also ensure equal access to key information across all faculty demographic groups.

INTERDISCIPLINARY/TEAM MENTORING

An added complexity to mentorship occurs when a faculty member works primarily in a center or institute, but academic milestones and advancement processes remain the responsibility of the department. Close collaboration between department and center/institute will be required to serve the needs of this group of faculty. For interdisciplinary mentoring to be effective, a few additional strategies should be utilized:

Define Roles and Team Structure

Clarification of the roles and structure of the mentoring group is essential for an effective interdisciplinary or team mentoring relationship. All who advise in this setting should know for what particular types of guidance they will be responsible. Boundaries, as well as the specific work for which each individual is responsible, should be clarified. While scholarly advising may be best done within the center, advice regarding promotion should come from the department that is responsible for executing this process.

Communication

Because the structure of interdisciplinary mentoring relationships are more complex than traditional departmental or school-based mentoring, effective communication is crucial to ensure that center/institute-based faculty are well advised with respect to scholarship, academic advancement, and network development. Thus, communication pathways between school/department and centers need to be clearly established.

ESTABLISHING OVERSIGHT, EVALUATION, AND SUSTAINABILITY

For systematic advising/mentoring efforts initiated and supported by a unit to be sustained and successful, they require both oversight and evaluation by the school/ department. Assurance that mentoring is occurring and assessment that junior faculty are satisfied with the quality and quantity of mentoring are evaluative tasks that can be done at different levels of complexity. Oversight may also be managed by a school/department designee, who can provide tools and resources, as well as consult those involved, for appraisals of needs and satisfaction with the program. Evaluation of program efficacy may be done using simple survey tools, faculty activity reports for evaluating the mentoring, teaching evaluations, and faculty advancement outcomes.

School or departmental plans that strongly rely on assigned pairing of faculty should also have a mechanism for "unpairing" should the relationship prove problematic for any reason. It is desirable for the unit to have some neutral and confidential way of evaluating and managing unsuccessful or difficult mentoring interactions.

Regardless of the model chosen, the departmental leadership should determine the priority of mentorship efforts within the departmental portfolio of work and ensure that a proper evaluation and oversight plan is included in their mentorship efforts.

As the definition of mentoring can vary greatly depending on the roles, models, and functions the individuals in the mentoring relationship perform, evaluating mentorship requires a consensus on the operational definition of mentoring and the specific outcomes most important to the academic unit from which to derive measurements and assessments of effectiveness and success (Berk, Berg, Mortimer, Walton-Moss, & Yeo, 2005). Mentoring programs and relationships grow and change over time, so it is important to use both formative and summative evaluation principles in evaluating mentorship initiatives.

Nevertheless, mentorship initiatives can be evaluated in the following areas:

- competencies and satisfaction of the mentor
- competencies and satisfaction of the mentee
- quality of the support provided by mentors
- achievement of defined goals and outcomes for mentoring
- quality of the program resources

PART III - TOOLS FOR MENTORS

In higher educational institutions, teaching and supervising students at all levels is coequal in importance to the generation of scholarship and is considered an essential measurable parameter guiding promotion decisions. By contrast, mentorship of junior colleagues by senior colleagues is often the product of an individual commitment by a senior faculty member relying on "natural skills and intuition." In addition, this type of mentoring relationship is occurring between colleagues, albeit at different career stages, but which influences the interactions. Mentoring skills, like teaching skills, can be enhanced by specific training and attentiveness to key principles and best practices (Pfund et al., 2006):

- Time Commitment: Because mentoring requires ongoing advising and review of a junior colleague's scholarship and teaching, senior faculty should be sure they can commit adequate time to this task.
- Collaboration: In contrast to teaching students, which
 may be supervisory in nature, mentoring junior colleagues
 requires a more collaborative working approach with the
 goal of scholarly independence and recognition as an
 expert in discipline.
- Scope of Guidance: Mentors and mentees, whether assigned by the academic unit or self-selected, should establish what the scope of mentoring will be. Will the focus be scholarship, teaching, overall career development, advancement, sponsorship, or combinations of these? It is particularly important, when there is a mentoring team or interdisciplinary mentoring is planned, that there is clarity regarding the team structure and individual responsibilities of team members.
- Mentoring Plan: Mentors should collaboratively work with mentees to agree upon the mentorship goals, frequency of meetings, who will create the agenda for any meetings, and the timetable for deliverables, as well as the duration of the relationship. In general, the areas of career development that should be discussed during meetings may include
 - o development of scholarship and progress toward scholarly independence;
 - o educational skills and evaluation;
 - o network development opportunities;
 - o preparation for and progress toward academic promotion; and
 - o advice on academic service (what, how much and when are key considerations).

- Special Considerations: Mentors should assess whether there are considerations that would be especially important for women or underrepresented minority faculty that should be discussed. If the mentor does not feel comfortable or sufficiently experienced to discuss such specific topics, facilitating meeting with other faculty members better able to address these concerns is recommended. Mentees should also feel empowered to broaden their group of advisors based on their specific concerns and their level of comfort in discussing these concerns.
- Record Keeping: Mentors and mentees should mutually decide what kind of record of meetings should be retained. It is advantageous to both the mentor and mentee to keep written records that allow measurement of progress and accomplishment of goals. The exact format will depend on the nature of the work and the wishes of mentors and mentees, as well as the departmental or school culture.
- Network Development: Mentors should encourage and facilitate, if possible, the development of professional and broadened mentoring networks. In addition to the professional growth opportunities, such networks can provide alternative or complementary career development perspectives. They may provide a comfortable environment to discuss some of the unique concerns of women and underrepresented minority faculty.

PART IV - STRATEGIES FOR MENTEES

Whether or not academic units have structured mentoring programs, or junior faculty have been fortunate enough to find more senior faculty willing to act informally as mentor, there are specific strategies and best practices that junior faculty may employ that will allow them to expand their opportunities for mentorship and maximize the benefit of such relationships. Because mentoring relationships are active collaborations, they require commitment and engagement, as well as ongoing self-reflection and evaluation on the part of junior faculty. Thus, it is essential that the faculty members seeking guidance clearly formulate career goals and define what they wish their key accomplishments to be.

A useful construct for faculty to organize a career plan is a brief written summary statement including an overarching vision of the impact on a field they wish to have, the specific areas (mission) in which they will work to realize that vision, and then the specific goals (strategic goals) with timelines for accomplishments that will mark progress within their field. Junior faculty should then consider what competencies they need and plan how and when these should be acquired. This construct provides a framework for discussion and development of specific strategic goals. Faculty should also recognize that the specifics of this construct may change over time and with mentoring advice, but that the initial framework provides a working document for reference. Once a framing construct is articulated, junior faculty should identify in what specific areas guidance is needed and who might best provide such guidance. They should be encouraged to think broadly regarding sources of career development advice and to develop a network including senior faculty, peers and near peers, as well as individuals sharing similar challenges or demographics. Networks should provide opportunities for reciprocal exchange of information relevant to career advancement but also for expressive or psychosocial support.

It is important for junior faculty to be aware of key domains of importance for their success (i.e., teaching; independent scholarship; professional networking; academic or institutional service) and to be sure that appropriate guidance is provided. They should also measure progress in each of these domains within the context of timelines for academic advancement:

- Expectations for achieving academic goals and timelines
- Setting appropriate academic goals and timelines for individual development of these goals
- Direction of scholarly work
- Development of the timelines and measurement of progress, i.e., the type of grants or other support for which to apply and expected scholarly output
- Perspectives and guidance for teaching; student supervisory efficacy
- Advising and preparation for promotion processes and timelines
- Advising on academic service/citizenship—what, how much, and when
- School or department-specific policies surrounding advancement
- Facilitating connection between potential collaborators
- Guidance for institutional navigation (school/department history, priorities, formal and informal rules, structures, practices, and resources)
- Annual discussion of scholarly work and goals as part of long-term goals
- Sponsorship opportunities (nomination for key organizational memberships, lectureships, honors, and awards when appropriate)
- Assistance with opportunities to build networks within the institution and externally within the scholarly field
- Identification of barriers or impediments to progress
- Development of solutions to overcome impediments
- Provision of constructive feedback and encouragement
- Identification of other individuals who may provide guidance on academic or nonacademic matters on which the mentor is less able to advise
- Identification of opportunities for collaboration

MENTORING (DEVELOPMENTAL) NETWORKS

Perhaps one of the most important conceptual changes regarding mentorship in both business and academia is the emphasis on the need for more than a single mentor (DeCastro et al., 2013a; Pololi & Knight, 2005; De Janasz & Sullivan, 2004; Van Emmerik, 2004; Ibarra, 1993). Multiple mentors who may provide both technical career advancing advice and support for psychosocial needs are important components of a mentoring network.

DeCastro et al. (2013a) surveyed 100 NIH research grant recipients and 28 research mentors regarding their mentorship experiences and found that the need for more than a single mentor emerged as a recurrent theme in interviews with both mentors and mentees. The reasons given by both types of participants included:

- 1. Identification by mentees of unique guidance needs
- 2. Additional sources of information from other perspectives
- 3. Significant technical, informational, and psychosocial support from peer groups
- 4. The importance of having some mentors based on demographic factors, i.e., gender congruence
- 5. Recognition that guidance needs might evolve over time and thus the network might evolve with time

A larger European study (Van Emmerik, 2004) focused on the specific characteristics that developmental networks associated with positive career success. The author found that the size of network, the range (the diversity of positions or roles of members within the network), the strength of the ties and longer duration of network relationships were network characteristics positively correlated with greater career success and satisfaction.

Ibarra (1993) specifically examined the characteristics of effective mentoring networks, directly comparing the networks of women and minorities to those of majority men. Of importance, she found that the networks of women and underrepresented minorities significantly differed qualitatively from those of majority men. Networks of women and minorities often provided more psychosocial support from similar individuals of similar rank, and thus less technical career advancing support, which comes more easily from individuals of higher organizational rank and status. Further, she makes the very clear point that "the

opportunity context" for network development may be a more important driver of limited network composition than the preferences of women and underrepresented minorities. Stated somewhat differently, it may not be the case that women and minorities prefer more psychosocial support, but rather that they lack the opportunities to develop networks that include individuals capable of providing technical support and sponsorship.

In considering developmental network formation, faculty should include individuals who are knowledgeable and supportive of technical career advancement, who may be peers or superiors. Other network members may be more supportive of psychosocial needs, and relationships may be based on similarity in goals or challenges. Women and underrepresented minorities should be mindful to be certain that the range and diversity of their networks support both technical and psychosocial aspects of career advancement.

APPENDIX A

CHECKLIST: BEST PRACTICES FOR SCHOOLS/ DEPARTMENTS, MENTORS, AND MENTEES

FOR SCHOOLS/DEPARTMENTS

	el faculty, underrepresented minority, etc.)		
☐ Clearly articulate career priorities and expectations for faculty that mentorship programs will address*	☐ Establish meeting parameters (i.e., frequency of meetings; content; record keeping)		
☐ Develop independent scholarship/research	☐ Determine tools and skills needed by mentors		
productivity	☐ Determine tools and skills needed by mentees		
☐ Develop important professional networks			
☐ Provide guidance regarding the determinants (both	☐ Identify and articulate roles and responsibilities ☐ Define specific expectations for each kind of mentor ing relationship ☐ Define and communicate individual and shared responsibilities of mentors and mentees		
formal and informal) essential for academic advance- ment in that unit			
☐ Support teaching performance			
☐ Proactively recognize and mitigate factors that dis-			
proportionately deter the advancement of women and underrepresented minority faculty	☐ Communicate expectations surrounding deliverables		
	and measures of progress for department, mentor,		
☐ Identify whether additional psychosocial aspects of mentoring will be addressed by departmentally facilitated programs	and mentee		
*School/departmental programs should aim, at minimum,	☐ Establishing Oversight, Evaluation, and Sustainability ☐ Identify metrics that the unit will use for evaluating the effectiveness and impact of mentoring programs and relationships (i.e., mentee/mentor satisfaction,		
to provide technical career support but may also facilitate psychosocial support			
☐ Identify mentoring models most feasible for the unit	specific scholarly product delivery, promotion)		
☐ Senior-junior dyads	☐ Establish methodology to assure that needs of men- tors and mentees are met		
☐ Group seminars	☐ Assess junior faculty satisfaction with the quality and		
☐ Peer or near peer mentoring	quantity of mentoring		
☐ Demographically based peer groups	\square Assess mentor satisfaction with preparation for and		
☐ Sponsorship	support of his or her role		
☐ Coaching	 Identify and provide tools and resources to ensure ongoing support 		
☐ Identify the structure and resources needed	$\hfill\square$ Determine tools and skills needed and responsibilities		
☐ Mentorship training for senior faculty	needed by mentees		
☐ Departmental reward for mentoring work	☐ Decide on a confidential process for mentors and mentees to voice concerns		
\square Curriculum and facilitators for group programs	☐ Decide on a confidential mechanism to resolve seri-		
\square Facilitation of peer, near peer, or group meetings and	ous differences between mentor and mentees		

demographically focused groups (i.e., women, midlev-

FOR MENTORS

- ☐ Time Commitment: Mentors should be able to commit to and honor the time required for meeting and advising their mentee(s).
- ☐ Skills and Needs: Mentors should assess their skills for mentorship and determine their developmental needs.
- □ Collaboration: Mentors should work with the mentee(s) on the development of realistic career goals and timelines for achieving those goals.
- ☐ Scope of Guidance: Mentors should be able to provide guidance on setting objectives, vision, and strategies for the specific scope of advising.
- Mentoring Plan: It is the responsibility of the mentee to provide the mentor with his or her goals, a CV, teaching portfolio, research statement, and whatever other materials will be useful in forming a mentorship plan; it is the responsibility of the mentor to collaborate with the mentee to develop attainable goals, types of guidance, and resources for developing necessary academic competencies, relationships, and measurement of progress.
- □ Communication: In order for mentorship to be most effective, both the mentee and the mentor must establish a level of trust with each other. Both should practice careful and active listening and be able to communicate respectfully and confidentially. Potential communication barriers based on race, ethnicity, culture, or background are important to consider. Mechanisms to provide constructive feedback should be thoughtfully considered.
- **Network Development:** A key role of a mentor is to help facilitate the development of academic networks.
- ☐ **Sponsorship:** Some faculty may benefit from sponsorship in addition to technical mentorship. Sponsors advocate for their mentees and use their influence to help a mentee's career advance.
- □ **Diversity in Mentorship:** Careful consideration of issues that may arise with mentorship across boundaries of gender/race/ethnicity/sexual orientation/culture/religion.

FOR MENTEES

- □ Identify Needs and Interests: Mentees should start by identifying their own needs and interests in order to create a draft of career goals and objectives. These needs and interests should be provided to the mentor, along with a draft of goals and any supporting materials (i.e., CV, research statement, teaching portfolio, individual development plan, etc.), prior to an initial meeting.
- □ Active Engagement: To ensure effective mentorship, it is essential for the mentee to be actively engaged in the mentoring relationship. This includes listening attentively, a willingness to work outside of one's "comfort zone," identifying specific developmental goals for which guidance is sought, initiation of meetings, adequate preparation for meetings, soliciting feedback and willingness to listen to feedback both positive and negative, and taking responsibility for developing scholarly independence.
- □ Identify Any Gaps in Competencies/Skill Sets: A responsible mentee will also reflect on their own activities and goals and identify gaps in competencies and/or skill sets that may be essential to the mentee's ability to successfully attain his or her goals.
- ☐ Setting Meeting Times and Agendas: It is the responsibility of the mentee to set meeting times and agendas with a mentor. Though the agendas may be set collaboratively, a successful mentoring relationship requires the mentees to be respectful of the time of his or her mentor(s) and to be prepared for meetings with questions, materials, and a commitment to work hard.
- ☐ **Developing Networks:** The mentee should be actively engaged in developing a broad network of developmental relationships.

APPENDIX B

LOGIC MODEL FOR SCHOOLS/ACADEMIC UNITS IMPLEMENTING MENTORING PROGRAMS

INPUTS What is invested Mentors

Faculty



Staff



Time



Resources



Materials



PROGRAM DESIGN

Mentoring Model:

- Hierarchical
- Peer or near peer
- Group
- Mentoring networks

Format:

- Individual or group
- Formal or informal
- Mentor/mentee pairing
- Interdisciplinary/team

Delivery:

- Frequency of meetings
- Duration of meetings
- Duration of program
- Record of meetings

Content:

- Informational
- Advisory
- Psychosocial support
- Appraisal and evaluation

Resources:

- Professional development activities
- Tools for communication or programs
- Annual evaluation
- Incentives for mentors and mentees

OUTCOMES - IMPACT

Short Term

Mentees:

- Enhanced knowledge and skills
- Increased satisfaction
- Clear sense of professional goals and timelines
- Increased confidence
- Enhanced productivity for professional outcomes (scholarship, grants, publications, education, overall citizenship)

Mentors:

- Increased job satisfaction
- Strengthening the department
- Investment in mentee's career as part of a legacy
- Increased collaboration
- Exposure to new ideas

Long Term

Mentees:

- Increased satisfaction
- Satisfactory productivity
- Establishing independence of scholarship
- External recognition

Mentors:

- Satisfaction with mentor's career
- Organizational recognition
- Increased retention

School/Department:

- Faculty success with scholarship
- Increased faculty promotions
- Increased faculty retention
- Increased hires
- Decreased faculty attrition

Alexander, H., & Lang, J. (2008). The long-term retention and attrition of US medical school faculty. *Analysis in Brief*, 8(4). Washington, DC: Association of American Medical Colleges.

Allen, T. D., Eby, L. T., Poteet, M. L., Lentz, E., & Lima, L. (2004). Career benefits associated with mentoring for protégés: A meta-analysis. *Journal of Applied Psychology*, 89(1), 127-136.

Berk, R. A., Berg, J., Mortimer, R., Walton-Moss, B., & Yeo, T. P. (2005). Measuring the effectiveness of faculty mentoring relationships. *Academic Medicine: Journal of the Association of American Medical Colleges*, 80(1), 66-71.

Bhagia, J., & Joyce A. (2000). The mentoring partnership. *Mayo Clinic Proceedings*, 75(5), 535–537.

Bickel, J. (2014). How men can excel as mentors of women. Academic Medicine: *Journal of the Association of American Medical Colleges*, 89(8), 1–3.

Bland, C. J., Center, B. A., Finstad, D. A., Risbey, K. R., & Staples, J. G. (2005). A theoretical, practical, predictive model of faculty and department research productivity. Academic Medicine: Journal of the Association of American Medical Colleges, 80(3), 225–237.

Bland, C. J., & Ruffin, M. T. (1992). Characteristics of a productive research environment: Literature review. *Academic Medicine*, 67, 385-397.

Bland, C. J., & Schmitz, C. C. (1986). Characteristics of the successful researcher and implications for faculty development. *Journal of Medical Education*, 61(1), 22–31.

Bland, C. J., Seaquist, E., Pacala, J. T., Center, B., & Finstad, D. (2002). One school's strategy to assess and improve the vitality of its faculty. *Academic Medicine*, 77, 368-76.

Bland, C. J., Taylor, A. L., Shollen, L. S., Weber-Main, A. M., & Mulcahy, P. A. (2009). *Faculty success through mentoring*. Lanham, MD: Rowman & Littlefield Education.

Bland, C. J., Taylor, A., & Shollenberger, S. (Summer, 2006). Mentoring systems: Benefits and challenges of diverse mentoring partnerships. *Faculty Vitae*. Washington, DC: AAMC.

Bland, C., Weber-Main, A., Lund, S., & Finstad, D. (2005). The research productive department: Strategies from departments that excel. Bolton, MA: Anker Publishing Co., Inc.

Boice, R. (1992). The new faculty member. San Francisco: Jossey-Bass.

Bower, D. J., Diehr, S., Morzinski, J., & Simpson, D. (1999). Mentoring guidebook for academic physicians. 2nd ed. Milwaukee: Center for Ambulatory Teaching Excellence, Medical College of Wisconsin.

Boyle, P., & Boice, B. (Spring, 1998) Systematic mentoring for new faculty teachers and graduate teaching assistants. *Innovative Higher Education*, 22(3), 157–179.

Bryant, A. L., Brody, A., Perez, A., Shillam, C., Edelman, L. S., Bond, S. M., Foster, V., & Siegel, E. O. (2015). Development and implementation of a peer mentoring program for early career gerontological faculty. *Journal of Nursing Scholarship*, 47(3), 258-266.

Byrne, M. W., & Keefe, M. R. (2002). Building research competence in nursing through mentoring. *Journal of Nursing Scholarship*, 34(4), 391–396.

Chao, G. T., Walz, P. M., & Gardner, P. D. (1992). Formal and informal mentorships: A comparison on mentoring functions and contrast with nonmentored counterparts. *Personnel Psychology*, 45(3), 619.

Chesler, N. C., & Chesler, M. A. (2002). Gender-informed mentoring strategies for women engineering scholars: On establishing a caring community. *Journal of Engineering Education*, 91(1), 49–55.

Chesler, N. C., Single, P. B., & Mikic, B. (2003). On Belay: Peer-mentoring and adventure education for women faculty in engineering. *Journal of Engineering Education*, 92(3), 257–262.

Corcoran, M., & Clark, S. M. (1984). Professional socialization and contemporary career attitudes of three faculty generations. *Research in Higher Education*, 20(2), 131–153.

Cropsey, K. L., Masho, S. W., Shiang, R., Sikka, V., Kornstein, S. G., & Hampton, C. L. (2008). Why do faculty leave? Reasons for attrition of women and minority faculty from a medical school: Four-year results. *Journal of Women's Health* (2002), 17(7), 1111–1118.

Daley, S., Wingard, D. L., & Reznik, V. (2006). Improving the retention of underrepresented minority faculty in academic medicine. *Journal of the National Medical Association*, 98(9), 1435–1440.

DeCastro, R., Sambuco, D., Ubel, P. A., Stewart, A., & Jagsi, R. (2013a). Mentor networks in academic medicine: Moving beyond dyadic conception of mentoring for junior faculty researchers. *Academic Medicine*, 88(4), 488-496.

DeCastro, R., Sambuco, D., Ubel, P. A., Stewart, A., & Jagsi, R. (2013b). Batting 300 is good: Perspectives of faculty researchers and their mentors on rejection, resilience, and persistence in academic medical careers. *Academic Medicine*, 88(4): 497–504.

De Janasz, S. C., & Sullivan, S. E. (2004). Multiple mentoring in academe: Developing the professorial network. *Journal of Vocational Behavior*, 64(2), 263–283.

De Janasz, S. C., Sullivan, S. E., Whiting, V., & Biech, E. (2003). Mentor networks and career success: Lessons for turbulent times. *Academy of Management* 17(4), 78–93.

Doran, G. T. (1981). There's a SMART way to write management's goals and objectives. *Management Review*, 70(11), 35-36.

Fang, D., Moy, E., Colburn, L., & Hurley, J. (2000). Racial and ethnic disparities in faculty promotion in academic medicine. *Journal of the American Medical Association*, 284(9), 1085–1092.

Fleming, G. M., Simmons, J. H., Xu, M., Gesell, S. B., Brown, R. F., Cutrer, W. B., Gigante, J., & Cooper, W. O. (2015). A facilitated peer mentoring program for junior faculty to promote professional development and peer networking. *Academic Medicine* 90(6), 819–826.

Fogg, P. (2005, May 27). Harvard committees suggest steps to help women. The Chronicle of Higher Education, The Faculty, (38), A8.

Fried, L. P., Francomano, C. A., MacDonald, S. M., Wagner, E. M., Stoke, E. J., Carbone, K. M., Bias, W. B., Newman, M. M., & Stobo, J. D. (1996) Career development for women in academic medicine: Multiple interventions in a department of medicine. *Journal of the American Medical Association*, 276(11), 898–905.

Gagliardi, A. R., Webster, F., Perrier, L., Bell, M., & Straus, S. (2014). Exploring mentorship as a strategy to build capacity for knowledge translation research and practice: A scoping systematic review. *Implementation Science* 9(122), doi:10.1186/s13012-014-0122-z.

Girves, J. E., Zepeda, Y., & Gwathmey, J. K. (2005). Mentoring in a post-affirmative action world. *Journal of Social Issues*, 61(3), 449-479.

Guevara, J. P., Adanga, E., Avakame, E., & Carthon, M. B. (2013). Minority faculty development programs and underrepresented minority faculty representation at US medical schools. *Journal of the American Medical Association*, 310(21), 2297–2304.

Hitchcock, M. A., Bland, C. J., Hekelman, F. P., & Blumenthal, M. G. (1995). Professional networks: The influence of colleagues on the academic success of faculty. *Academic Medicine*, 70(12), 1108–1116.

Hsieh, E., Nunez-Smith, M., & Henrich, J. B. (2013). Needs and priorities in women's health training: Perspectives from an internal medicine residency program. *Journal of Women's Health* (Larchmt), 22(8), 667–672.

Ibarra, H. (1993). Personal networks of women and minorities in management: A conceptual framework. *Academy of Management Review*, 18(1), 56-87.

Ibarra, H., Carter, N. M., & Silva, C. (September, 2010). Why men still get more promotions than women. *Harvard Business Review*, 88(9), 80-126.

Ibarra, H., Ely, R., & Kolb, D. (September, 2013). Women rising: The unseen barriers. *Harvard Business Review*, 91(9), 81–95.

Ibarra, H., & Hunter, M. (2007). How leaders create and use networks. *Harvard Business Review*, 85(1), 62-70.

Jackson, V. A., Palepu, A., Szalacha, L., Caswell, C., Carr, P. L., & Inui, T. (2003). "Having the right chemistry": A qualitative study of mentoring in academic medicine. Academic Medicine: Journal of the Association of American Medical Colleges, 78(3), 328–334.

Johnsrud, L. K., & Sadao, K. C. (1998). The common experience of "otherness": Ethnic and racial minority faculty. *The Review of Higher Education*, 21(4), 315–342.

Kram, K. E. (1988). Mentoring at work: Developmental relationships in organizational life. University Press of America.

Kram, K. E., & Isabella, L. A. (1985). Mentoring alternatives: The role of peer relationships in career development. *Academy of Management Journal*, 28(1), 110-132.

Law, A. V., Bottenberg, M. M., Brozick, A. H., Currie, J. D., DiVall, M. V., Haines, S. T., ... Yablonski, E. (2014). A checklist for the development of faculty mentorship programs. *American Journal of Pharmaceutical Education*, 78(5), 98.

Liu, C. Q., & Alexander, H. (2010). Promotion rates for first-time assistant and associate professors appointed from 1967 to 1997. *Analysis in Brief*, 9(7). Washington, DC: Association of American Medical Colleges.

Mahoney, M. R., Wilson, E., Odom, K. L., Flowers, L., & Adler, S. R. (2008). Minority faculty voices on diversity in academic medicine: Perspectives from one school. Academic Medicine: Journal of the Association of American Medical Colleges, 83(8), 781–786.

Mark, S., Link, H., Morahan, P. S., Pololi, L., Reznik, V., & Tropez-Sims, S. (2001). Innovative mentoring programs to promote gender equity in academic medicine. Academic Medicine: Journal of the Association of American Medical Colleges, 76(1), 39-42.

McClurken, J. (October 14, 2009). Why not to set up a formal new faculty mentoring program. *The Chronicle of Higher Education*, 1–3.

McMurtie, B. (July 21, 2014). The difference mentoring makes. The Chronicle of Higher Education, 1-7.

Melicher, R. (2000). The perceived value of research and teaching mentoring by finance academicians. *Financial Practice and Education*, 10, 166–174.

Mott, V. (2002). Emerging perspectives on mentoring: Fostering adult learning and development. In C. A. Hansman (ed)., *Critical Perspectives on Mentoring: Trends and Issues.* ERIC. ED-465-045.

Nunez-Smith, M. (2007). Impact of race on the professional lives of physicians of African descent. Annals of Internal Medicine, 146(1), 45.

Nunez-Smith, M., Pilgrim, N., Wynia, M., Desai, M. M., Bright, C., Krumholz, H. M., & Bradley, E. H. (2009). Health care workplace discrimination and physician turnover. *Journal of the American Medical Association*, 101(12), 1274–1282.

Nunez-Smith, M., Pilgrim, N., Wynia, M., Desai, M. M., Jones, B. A., Bright, C., ... Bradley, E. H. (2009). Racel ethnicity and workplace discrimination: Results of a national survey of physicians. *Journal of General Internal Medicine*, 24(11), 1198-1204.

Palepu, A., Carr, P. L., Friedman, R. H., Amos, H., Ash, A. S., & Moskowitz, M. A. (1998). Minority faculty and academic rank in medicine. *Journal of the American Medical Association*, 280(9), 767–771.

Patton, L. D. (2009). My sister's keeper: A qualitative examination of mentoring experiences among African American women in graduate and professional schools. *The Journal of Higher Education*, 80(5), 510–537.

REFERENCES

Pfund, C., Maidl Pribbenow, C., Branchaw, J., Miller Lauffer, S., & Handelsman, J. (2006) Professional skills: The merits of training mentors. *Science*. 311(5760), 473-474.

Pololi, L. H., Civian, J. T., Brennan, R. T., Dottolo, A. L., & Krupat, E. (2013). Experiencing the culture of academic medicine: Gender matters, a national study. *Journal of General Internal Medicine*, 28(2), 201–207.

Pololi, L., Clay, M. C., Lipkin Jr., M., Hewson, M., Kaplan, C., & Frankel, R. M. (2001). Reflections on integrating theories of adult education into a medical school faculty development course. *Medical Teacher*, 23(3), 276–283.

Pololi, L., & Knight, S. (2005). Mentoring faculty in academic medicine. A new paradigm? *Journal of General Internal Medicine*, 20(9), 866–870.

Pololi, L. H., Knight, S. M., Dennis, K., & Frankel, R. M. (2002). Helping medical school faculty realize their dreams: An innovative, collaborative mentoring program. *Academic Medicine: Journal of the Association of American Medical Colleges*, 77(5), 377–384.

Reede, J. Y. (2014). Minority faculty development programs at US medical schools. *The Journal of the American Medical Association*, 311(11), 1157–1158.

Ross, J. S., Nunez-Smith, M., Forsyth, B. A., & Rosenbaum, J. R. (2008). Racial and ethnic differences in personal cervical cancer screening amongst post-graduate physicians: Results from a cross-sectional survey. *BMC Public Health*, 8(1), 378.

Sambuco, D., Dabrowsky, A., DeCastro, R., Stewart, A., Ubel, P. A., & Jagsi, R. (2013). Negotiation in academic medicine: narratives of faculty researchers and their mentors. *Academic Medicine* 88(4), 505-511.

Sandberg, S. (2013). Lean in: Women, work, and the will to lead. New York, NY: Alfred A. Knopf.

Sands, R., Parson, L., & Duane, J. (1991). Faculty mentoring faculty in a public university. *The Journal of Higher Education*, 62(2), 174–193.

Smith, K. A., Arlotta, P., Watt, F. M., & Solomon, S. L. (2015). Seven actionable strategies for advancing women in science, engineering, and medicine. *Cell Stem Cell*, 16(3), 221–224.

Snyder, T. D., & Dillow, S. A. (2015). Digest of education statistics 2013. National Center for Education, Institute of Education Sciences. United States Department of Education Statistics, (NCES 2012-001), Chapter 3.

Sorcinelli, M. D., & Yun, J. (2007). From mentor to mentoring networks: Mentoring in the new academy. *Change: The Magazine of Higher Learning*, 39(6), 58-61.

Sorcinelli, M. D., & Yun, J. H. (2009). When mentoring is the medium: Lessons learned from a faculty development initiative. *To improve the academy*, 27, 365-384.

Stanley, C. A., & Lincoln, Y. S. (2005). Cross-race faculty mentoring. *Change: The Magazine of Higher Learning*, 37(2), 44–50.

Tenner, E. (2004). The pitfalls of academic mentorships. *Chronicle of Higher Education*, 50(49), B7-B10.

Trix, F., & Psenka, C. (2003). Exploring the color of glass: Letters of recommendation for female and male medical faculty. *Discourse* & *Society*, 14(2), 191–220. doi:10.1177/0957926503014002277.

Turner, C. S. V., Myers Jr., S. L., & Creswell, J. W. (1999). Exploring underrepresentation: The case of faculty of color in the midwest. *Journal of Higher Education*, 70(1), 27-59.

Van Emmerik, I. J. (2004). The more you can get the better: Mentoring constellations and intrinsic career success. *Career Development International*, 9(6), 578–594.

Williams, J. C., Phillips, K., & Hall, E. (2014). *Double jeopardy? Gender bias against women of color in science*. Published online at www.worklifelaw.org.

Williams, L. S. (1991). The effects of a comprehensive teaching assistant training program on teaching anxiety and effectiveness. Research in Higher Education, 32(5), 585–598.

Yeo, H., Viola, K., Berg, D., Lin, Z., Nunez-Smith, M., Cammann, C., ... Curry, L. A. (2009). Attitudes, training experiences, and professional expectations of US general surgery residents: A national survey. *Journal of the American Medical Association*, 302(12), 1301–1308.

Zachary, L. (2000). The mentor's guide: Facilitating effective learning relationships. San Francisco, CA: Jossey-Bass.

Zellers, D. F., Howard, V. M., & Barcic, M. A. (2008). Faculty mentoring programs: Reenvisioning rather than reinventing the wheel. *Review of Educational Research*, 78(3), 552–588.

Zhuge, Y., Kaufman, J., Simeone, D. M., Chen, H., & Velazquez, O. C. (2011). Is there still a glass ceiling for women in academic surgery? *Annals of Surgery*, 253(4), 637–643.